Author’s response to reviews

Title: Usefulness of dermoscopy for detecting skin leiomyomas in patient with uterine fibroids and cerebral cavernomas

Authors:

Laura Diluvio (lauradiluvio@yahoo.it)
Claudia Torti (torti@uniroma2.it)
Alessandro Terrinoni (terrinoni@uniroma2.it)
Eleonora Candi (candi@uniroma2.it)
Raffaella Piancatelli (piancatelli@uniroma2.it)
Emilio Piccione (piccione@uniroma2.it)
Evelin Jasmine Paternò (evelin.j@tiscali.it)
Sergio Chimenti (chimenti@uniroma2.it)
Augusto Orlandi (orlandi@uniroma2.it)
Elena Campione (campioneeleena@hotmail.com)
Luca Bianchi (luca.bianchi@unitoma2.it)

Version: 2 Date: 7 January 2014

Author’s response to reviews:

e-mails of each author were added
the contribution of each author was at the end of the manuscript;
no acknowledgments are given.